Title: Frankincense essential oil prepared from hydrodistillation of Boswellia sacra gum resins induces human pancreatic cancer cell death in cultures and in a xenograft murine model

Authors:

Xiao Ni (nixiao205122@126.com)
Mahmoud M Suhail (mahmoudsuhail@gmail.com)
Qing Yang (qing-yang@ouhsc.edu)
Amy Cao (amycao@usc.edu)
Kar-Ming Fung (karming-fung@ouhsc.edu)
Russell G Postier (russell-postier@ouhsc.edu)
Cole Woolley (cwoolley@youngliving.com)
Gary Young (sundance@youngliving.com)
Jingzhe Zhang (zjzzzq@sina.com)
Hsueh-Kung Lin (hk-lin@ouhsc.edu)

Version: 4 Date: 6 December 2012

Author's response to reviews:

Dear Dr. Rowles:

Please find attached re-revised version of the submitted manuscript. We have corrected two typographical errors identified by the reviewer. We sincerely hope that the revised manuscript is acceptable to be published in BMC Complementary and Alternative Medicine.

Sincerely yours,

HK Lin